Author's response to reviews

Title: Building bridges to patients: design and pilot evaluation of a training session in argumentation theory for chronic pain experts

Authors:

Claudia Zanini (claudia.zanini@paraplegie.ch)
Piercarlo Sarzi-Puttini (piercarlo.sarziputtini@gmail.com)
Fabiola Atzeni (atzenifabiola@hotmail.com)
Manuela Di Franco (Manuela.Difranco@uniroma1.it)
Sara Rubinelli (sara.rubinelli@paraplegie.ch)

Version: 2 Date: 3 April 2014

Author's response to reviews:

As requested from the Journal Editorial Office, I am submitting a revised version of our manuscript entitled “Building bridges to patients. Design and evaluation of a training session in argumentation theory for chronic pain experts” (MS ID: 1165041359125367) for consideration for publication in BMC Medical Education.

Here is a point-by-point description of the changes we made:
1. We included the email addresses of all authors (in the Title page).
2. We included in the manuscript a statement, which attests that we received ethical approval for our study from the regional ethics committee, and that informed consent to participate in the study was obtained from all participants (in the Methods section).
3. We included in the manuscript a declaration of non-competing interests (between the Conclusion and the Authors contribution sections).

Thank you for receiving our manuscript and proceeding to editorial assessment.

Kind regards,
Claudia Zanini